Author’s response to reviews

Title: Effects of pediatric first aid training on preschool teachers: A longitudinal cohort study in China

Authors:

Feng Li (lifeng0228@163.com)
Xiaoyang Sheng (shengxiaoyangcn@aliyun.com)
Jinsong Zhang (zhangisk@gmail.com)
Fan Jiang (fanjjiang@shsmu.edu.cn)
Xiaoming Shen (xiaomingshen163@163.com)

Version: 6 Date: 13 August 2014

Author’s response to reviews: see over
RE: 1208510047118014

Dear Editors,

Thank you for your useful comments in our manuscript. We have modified the manuscript accordingly, and detailed corrections are listed below:

1. Author Emails

Thank you for uploading the completed STROBE checklist. Unfortunately author Shen now has two different email addresses xiaomingshen163@163.com and xmshen@shsmu.edu.cn. Please ensure that all author emails are identical on both the authors details page of the submission system and on the title page of the manuscript.

Answer: We have conformed and corrected this. Thanks.

2. Please also indicate in the methods section of your manuscript that consent was obtained from the participants.

Answer: We have supplemented this in the manuscript. Thanks.

We look forward to hearing from you and thank for your consideration.

Best regards,

Feng Li, Xiaoyang Sheng, Jinsong Zhang, Fan Jiang, Xiaoming Shen